Author's response to reviews

Title: Prevalence of sexually transmitted infections among pregnant women with known HIV status in northern Tanzania.

Authors:

Sia E Msuya (siamsuya@hotmail.com)
Jacqueline Uriyo (jackieuriyo@yahoo.com)
Akhtar Hussain (akhtar.hussain@medisin.uio.no)
Elizabeth M Mbizvo (embizvo@mweb.co.zw)
Stig Jeansson (s.l.jeansson@medisin.uio.no)
Noel E Sam (noelsam@kcmc.ac.tz)
Babill Stray-Pedersen (babill.stray-pedersen@medisin.uio.no)

Version: 3 Date: 16 December 2008

Author's response to reviews:

QUERIES

K 1: Add "statistically significant"

K 2: We meant both pregnant women and the infants received the single dose of Nevirapine for prevention of mother-to-child transmission of HIV. Women take the tablet at the onset of labour, while their infants are given the Nevirapine syrup within 72 hours after delivery.

K 3: It is “all” the women. Here by all women we meant HIV-positive and HIV-negative women

K 4: potassium hydroxide (KOH) preparations

K 5: It is “trichomoniasis” (TV)

K 6: It is small letter “chlamydia”.